UCLA UCLA Previously Published Works

Title

Discontinuation and non-publication of heart failure randomized controlled trials: a call to publish all trial results

Permalink https://escholarship.org/uc/item/6mj172q3

Journal ESC Heart Failure, 8(1)

ISSN 2055-5822

Authors

Khan, Muhammad Shahzeb Shahid, Izza Asad, Nava <u>et al.</u>

Publication Date

2021-02-01

DOI

10.1002/ehf2.13099

Peer reviewed

Discontinuation and non-publication of heart failure randomized controlled trials: a call to publish all trial results

Muhammad Shahzeb Khan¹, Izza Shahid², Nava Asad³, Stephen J. Greene^{4,5}, Safi U. Khan⁶, Rami Doukky⁷, Marco Metra⁸, Stefan D. Anker⁹, Gerasimos S. Filippatos^{10,11}, Gregg C. Fonarow¹² and Javed Butler^{13*}

¹Department of Medicine, Cook County Health Sciences, Chicago, IL, USA; ²Department of Medicine, Ziauddin Medical University, Karachi, Pakistan; ³Department of Medicine, Dow University of Health Sciences, Karachi, Pakistan; ⁴Division of Cardiology, Duke University School of Medicine, Durham, NC, USA; ⁵Duke Clinical Research Institute, Durham, NC, USA; ⁶Department of Medicine, West Virginia University, Morgantown, WV, USA; ⁷Division of Cardiology, Rush University Medical Center, Chicago, IL, USA; ⁸Institute of Cardiology, ASST Spedali Civili, Department of Medicial and Surgical Specialties, Radiological Sciences, and Public Health, University of Brescia, Brescia, Italy; ⁹Department of Cardiology (CVK) and Berlin Institute of Health Center for Regenerative Therapies (BCRT), German Centre for Cardiovascular Research (DZHK) partner site Berlin, Charité Universiti Berlin, Berlin, Germany; ¹⁰Medical School, University of Cyprus, Nicosia, Cyprus; ¹¹National and Kapodistrian University of Athens, Athens University Hospital Attikon, Athens, Greece; ¹²Division of Cardiology, Ronal Reagan-UCLA Medical Center, CA, Los Angeles, USA; ¹³Department of Medicine, University of Medical Center, 2500 N. State Street, Jackson, MS 39216, USA

Abstract

Aims Discontinuation or non-publication of trials may hinder scientific progress and violates the commitment made to research participants. We sought to identify the prevalence of discontinuation and non-publication of heart failure (HF) clinical trials.

Methods and results We conducted a cross-sectional search of ClinicalTrials.gov to identify all completed and discontinued HF clinical trials. We limited our search to only include trials that were completed by 31 December 2017. Trials were investigated to identify reasons for discontinuation. Informative termination was defined as trial termination due to safety or efficacy concerns. Data pertaining to the trial phase, funding, intervention, enrolment, and trial completion date were extracted for each trial. A total of 572 trials were included. Of these, 21% (n = 118) were discontinued before completion. Patient accrual was the most frequently cited reason (n = 42; 36%) for trial discontinuation, followed by informative termination (n = 16; 14%) and funding (n = 14; 12%). Overall, 24 780 patients were enrolled in trials that were terminated. Of trials that were completed and not terminated, nearly one-third (n = 131/454; 29%) were not published. Seventy-nine (24%) trials were published within 12 months, 192 (59%) within 24 months, and 252 (78%) trials within 36 months.

Conclusions Discontinuation and non-publication of HF trials is common. This raises ethical concerns towards participants who volunteer for research and are exposed to potential risks, inconvenience, and discomfort without furthering scientific progress.

Keywords Heart failure; Clinical trials; Discontinuation; Non-publication; Enrolment

Received: 14 July 2020; Revised: 17 September 2020; Accepted: 22 October 2020

*Correspondence to: Javed Butler, Department of Medicine, University of Mississippi Medical Center, 2500 N. State Street, Jackson, MS 39216, USA. Tel: 601 984-5600; Fax: 601 984-5608. Email: jbutler4@umc.edu

Introduction

Randomized clinical trials (RCTs) provide the highest level of evidence to confirm treatment effects and inform clinical care. The success of conducting clinical trials depends upon the willingness of human subjects who volunteer to participate in these trials, often without any prior knowledge of the benefits and risks of unproven interventions.¹ It is vital for investigators and sponsors to conduct trials with the highest ethical standards² and publish detailed trial results, regardless of their outcome. Failure to do so can hinder scientific progress and neglects the commitment from trial participants who expose themselves to potential risks, inconvenience, and discomfort for the sake of generating medical evidence.^{3–5} Moreover, failure to publish and disseminate trial results leaves other patients and investigators with incomplete information that can lead to similar or duplicative studies being carried out or similar trial mistakes being

© 2020 The Authors. ESC Heart Failure published by John Wiley & Sons Ltd on behalf of European Society of Cardiology

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

made due to general unawareness of earlier results. This is especially true for drug trials that are stopped due to safety reasons as drugs of the same or similar class may pose the same safety concerns in future.

The World Health Organization (WHO) recently mandated that the main findings of all clinical trials must be posted to a primary clinical trial registry within 12 months of the end of the study and called for reporting of older unpublished trials.⁶ Similarly, the Food and Drug Administration Amendments Act of 2007 (FDAAA) also requires the public sharing of clinical trial results of FDA regulated drugs and devices to the ClinicalTrials.gov databank.⁷ However, despite the moral and ethical responsibility to publish studies, premature discontinuation of trials with no safety or futility concerns and non-publication of clinical trials remain prevalent across a range of specialties.⁸⁻¹³ Evidence suggests that trials with positive outcomes are published more frequently as compared with those with neutral outcomes.^{14,15} This may be of particular significance in trials that are slower to enrol, for example, heart failure (HF).¹⁶ We therefore aimed to investigate the discontinuation and non-publication of HF trials and subsequently determine the frequency of available results for unpublished trials. We also highlight possible strategies to enhance timely publication of trial results.

Methods

Data source and search strategy

ClinicalTrials.gov is the most comprehensive publicly available clinical trial registry.¹⁷ Investigators are expected to continuously update information as the trials progress in order to have the most up-to-date data available as public record.¹⁸ A cross-sectional search of trials for HF registered in the ClinicalTrials.gov database was conducted. The initial search was carried out in January 2020 by two independent investigators (N. A. and I. S.). We used the 'advance search' option to search for all randomized HF trials that were completed or had been discontinued since the inception of the database. Because the estimated median time to publication is 20 months, ¹⁹ the search was limited to only include trials that were completed by 31 December 2017. Trials were included if the recruitment status was listed as 'completed' or 'active, not recruiting' to ensure that no additional participants would be enrolled in the study. Studies with and without available results were included. Trials that were registered ≥30 days after the initial start date were excluded to avoid any possible bias by the investigator to only register the study due to positive study results.9

Data regarding trial phase, funding source, intervention, enrolment, and trial completion date were extracted from ClinicalTrials.gov for each study. Trials were grouped from phase I to phase IV. Trials described as phase I/II in their registry were categorized as phase II, while trials described as phase II/III in their registry were categorized as phase III. Funding source was further divided into industry funded, National Institute of Health (NIH), non-NIH US federal funding, or other (non-US funding, individuals, universities, or organizations), while trial intervention was further categorized as drug, behavioural, device, or procedural intervention.

For trials that were discontinued, reasons for trial discontinuation were analysed based on data provided at ClinicalTrials.gov. These included funding issue, loss of principal investigator, company/business decision, informative termination (defined as trial termination due to safety concerns, emerging efficacy data or short-term results indicating efficacy, or futility), slow enrolment, trial conduct problems (defined as technical or logistical issues contributing to lack of trial completion), or unreported/unclear reasons.

Publication search

ClinicalTrials.gov uses a unique NCT identifier number that automatically links to relevant PubMed publication. If a publication containing trial results was not listed, we searched Medline using NCT number, trial title, author names, study intervention, number of participants, study keywords, and funding sources to identify any existing trial publication. Publications search was carried out by two independent investigators (N. A. and I. S.) in January 2020. For those trials for which no published results were found, we repeated the search strategy on EMBASE. We considered a trial published if trial results were available in a peer-reviewed manuscript. A trial was considered unpublished if no relevant publication was identified in a peer-reviewed manuscript; these were categorized into those with or without results available in the ClinicalTrials.gov. Abstracts were not considered as peer reviewed publications.

Statistical analysis

Trials were categorized as either discontinued or completed. Continuous variables were reported as median (interquartile range [IQR]), while categorical variables were expressed as No. (%). All trial characteristics were calculated as a percentage of total trials. Time taken to publish studies was calculated from the trial completion date as reported by ClinicalTrials.gov to the date of first online or in print publication (whichever occurred first); χ^2 tests were used to evaluate associations between trial characteristics and trial completion and publication status. Multivariable logistic regression models assessing factors including funding source, interventions, trial phase, and masking were used to determine their impact on trial completion and publication, with completion or publication status as the dependent variable

and the control variables modelled as a set of dummy variables. Because trial publications can be late, a sensitivity analysis excluding trials completed after 2015 was also conducted to observe any meaningful change in trial statistics. All statistical analyses were performed using Statistical Package for the Social Sciences (SPSS) software version 23 (International Business Machines Corporation, New York, USA). All comparisons were two-tailed, and a *P-value <0.05* was considered statistically significant.

Results

Literature search

The initial search of the ClinicalTrials.gov database yielded 2954 HF trials, of which 572 were included in our analyses.

Main reasons for exclusion included recruitment status other than 'completed or active, not recruiting' (n = 1163), trial completion date after 31 December 2017 (n = 444), trial not randomized (n = 51), and trials that were registered with ClinicalTrials.gov more than 1 month after beginning enrolment (n = 724). Figure 1 highlights the detailed literature search process.

Trial characteristics

A total of 165 308 patients were enrolled in the 572 trials. The median number of participants enrolled across trials was 60 (IQR = 21–197). Nearly half (49%, n = 279) of trials studied drug interventions, while device, procedural, and behavioural interventions accounted for 23% (n = 134), 8% (n = 47) and 8% (n = 46) of trials, respectively. Forty-three per cent (n = 208) of trials were either phase 2 or phase 3.

FIGURE 1 Flowchart of the detailed literature search. Of the 572 included HF trials, 118 were discontinued. Of the 454 completed HF trials, 131 remain unpublished.



Almost half of the trials (47%, n = 270) received industry funding (*Table 1*). Enrolment of participants varied across trials; 258 trials (45%) enrolled less than 50 participants while 65 (11%) trials enrolled more than 500 participants, and 147 trials (26%) were completed before 2012, while 249 trials (44%) were listed as complete between the years 2015 and 2017. Twenty-three trials (4%) were for HF with preserved ejection fraction, and 49 trials (9%) were for acute heart failure. Seventy-four trials (13%) were double blind, while 247 trials (43%) were open label.

Discontinued trials

Of the 572 HF trials, 118 (21%) were discontinued before completion. Overall, 24 780 patients were enrolled in these trials at termination, and of these, 2849 (11.5%) were enrolled in trials that were discontinued based on informative reasons. There was a significant association between completion status and enrolment size of the trials (P < 0.01) (*Table 1*). Trials based on behavioural interventions were less likely to be completed compared with drug trials (OR 0.24, 95% CI [0.07, 0.79]; P = 0.02) (*Table 2*). On the contrary, device trials were significantly more likely to reach completion (OR 1.72, 95% CI [1.10, 2.70]; P = 0.02). Trials including triple blind

masking were also associated with increased odds of reaching trial completion compared with open label masking (OR 1.88 [1.01, 3.50]; P = 0.04).

The median number of participants enrolled in discontinued trials was 26 (IQR = 9–86) (*Table 1*). Slow enrolment was the most frequently cited reason (n = 42;36%) for trial discontinuation, while informative termination and funding issue accounted for 14% (n = 16) and 12% (n = 14) of the trials discontinued, respectively. Unreported or unclear reasons accorded for 22% (n = 26) of the discontinued trials (*Figure 2*). Majority of the trials discontinued were industry funded (n = 72; 61%), followed by funding by other organizations (n = 67; 56%), NIH funded trials (n = 10; 8%), and US Federal funding (n = 3; 2%). Of the 118 trials that were discontinued prematurely, 38 (32%) trials started before the year 2012, 20 (17%) between 2013 and 2014, while 18 (15%) trials commenced 2015 onwards, respectively.

Non-published trials

Of the 572 trials included, 454 (79%) were completed. Of the completed trials, 131 (29%) were unpublished (*Figure 1*). The median duration of time since trial completion was 55 months (IQR = 39-88). These 131 unpublished trials represented a

Table 1 Characteristics of discontinued randomized trials registered with ClinicalTrials.gov

Characteristics	All trials (n = 572), n (%)	Completed trials ($n = 454$), n (%)	Discontinued trials ($n = 118$), n (%)	P-value
Funding source ^a				0.063
Industry	270 (47)	198 (44)	72 (61)	
NIH	61 (11)	52 (11)	10 (8)	
US Federal	20 (3)	18 (4)	3 (2)	
Other (non-US)	352 (62)	286 (63)	67 (56)	
Intervention ^a				
Drug	279 (49)	212 (47)	67 (56)	0.050
Behavioural	47 (8)	44 (10)	4 (3)	
Device	134 (23)	98 (22)	37 (31)	
Procedure	46 (8)	39 (9)	8 (6)	
Other	125 (22)	102 (22)	24 (20)	
Trial phase ^b				
Phase 1	31 (5)	28 (6)	3 (3)	0.110
Phase 2	139 (24)	107 (24)	32 (27)	
Phase 3	109 (19)	81 (18)	27 (23)	
Phase 4	62 (11)	46 (10)	17 (14)	
Not applicable	231 (40)	192 (42)	39 (33)	
Masking				
Open label	247 (43)	203 (45)	45 (38)	0.16
Single blind	76 (13)	57 (13)	19 (16)	
Double blind	74 (13)	61 (13)	12 (10)	
Triple blind	58 (10)	39 (9)	19 (16)	
Quadruple blind	113 (20)	91 (20)	22 (19)	
Unspecified	4 (1)	3 (1)	1 (1)	
Enrolment size				
Median enrolment (IQR)	60 (21–197.25)	62 (29.25–220.75)	26 (9–86)	
SD no. of participants	869	839	625	
<10	53 (9)	22 (5)	31 (26)	< 0.01
<50	205 (36)	159 (35)	46 (39)	
<100	105 (18)	94 (21)	12 (10)	
<500	144 (25)	126 (28)	18 (14)	
>500	65 (11)	53 (12)	11 (9)	

NIH, National Institutes of Health.

^aTrials with multiple funding sources or interventions are listed within all relevant categories; totals therefore add to more than 100%. ^bTrials described as 'phase I/II' in their registry entry are categorized as phase II. Trials described as 'phase II/III' in their registry entry are categorized as phase II.

	Completion of trial		Publication of t	trial
	OR (95% CI)	P-value	OR (95% CI)	<i>P</i> -value
Funding source				
Industry	Reference		Reference	1
NIH	1.21 (0.44, 3.3)	0.71	1.53 (0.57, 4.14)	0.40
US Federal	0.34 (0.11, 1.04)	0.06	1.57 (0.60, 4.10)	0.36
Other	0.84 (0.55, 1.23)	0.43	0.54 (0.35, 0.83)	0.01
Intervention				
Drug	Reference		Reference	1
Behavioural	0.24 (0.07, 0.79)	0.02	0.78 (0.37, 1.67)	0.53
Device	1.72 (1.10, 2.70)	0.02	1.41 (0.85, 2.33)	0.18
Procedure	0.59 (0.26, 1.38)	0.22	1.15 (0.56, 2.38)	0.71
Other	1.18 (0.73, 1.91)	0.51	1.04 (0.62, 1.72)	0.89
Trial phase				
Phase 1	Reference		Reference	2
Phase 2	1.25 (0.47, 3.3)	0.66	4.21 (1.76, 10.10)	60;0.01
Phase 3	1.44 (0.54, 3.88)	0.47	4.41 (1.77, 11.01)	60;0.01
Phase 4	1.45 (0.50, 4.17)	0.49	2.39 (0.91, 6.26)	0.08
Not applicable	1.18 (0.46, 3.03)	0.73	2.48 (1.11, 5.53)	0.03
Masking				
Open label	Reference		Reference	2
Single blind	1.28 (0.71, 2.31)	0.42	0.79 (0.42, 1.51)	0.48
Double blind	0.83 (0.43, 1.61)	0.83	0.71 (0.37, 1.34)	0.29
Triple blind	1.88 (1.01, 3.50)	0.04	0.52 (0.23, 1.20)	0.13
Quadruple blind	0.91 (0.53, 1.59)	0.75	0.65 (0.37, 1.14)	0.13
Unspecified	3.57 (0.49, 25.96)	0.21	4.06 (0.36, 45.58)	0.26

Table 2 Logistic regression analysis of factors associated with completion and publication of heart failure clinical trials

OR, odds ratio.

FIGURE 2 The reasons for discontinuation of trials. Patient accrual was the most common cited reason for trial discontinuation.



total number of 18 471 participants. Thirty-three (25%) of these trials had more than 100 patients enrolled, and 32 trials (24%) were placebo controlled. Median number of participants enrolled in these unpublished trials was 43 (IQR = 20-100) (Table 3). Majority of these trials were based on drug interventions (n = 59; 45%), followed by device interventions (n = 34; 26%). A significant association was observed between trial publication status and funding source, trial phase, and enrolment size (P < 0.01). Thirty of the 131 (22.9%) unpublished trials comprising of 4887 participants posted results on ClinicalTrials.gov. None of these trials provided information on mortality during study time or adverse events. Seventy-six (58%) unpublished trials were completed by 2016. Sensitivity analysis excluding trials completed after 2015 revealed a total of 314 completed trials, of which 24.2% (n = 76) remain unpublished. Similar trends of trial characteristics were observed, with industry funded trials (55%) and trials based on drug interventions (46%) accounting for the highest percentage of unpublished trials.

Published studies

The median number of participants enrolled in trials that were subsequently published was 80 (IQR = 33-262) (Table 3), while the median time from study completion to publication was 559 (IQR = 351-887) days. Seventy-nine (24%) trials were published within 12 months, 192 (59%) trials within 24 months, and 252 (78%) trials within 36 months of trial completion date. Other sources of funding including non-US funding, individual funding, universities, or organizational funding had increased odds of trial publication compared with industry funded trials (OR 0.54, 95% CI [0.35, 0.83]; P = 0.01). Similarly, phase 2 (OR 4.21 95% CI [1.76, 10.10]; P < 0.01) and phase 3 trials (OR 4.41 95% CI [1.77, 11.01]: P < 0.01) were more likely to be published compared with phase 1 trials (Table 2). Trials that were published within 24 months compared with other published trials did not have any statistically significant difference in terms of sample size and industry funding.

Trial characteristics	All trials (n = 454), n (%)	Published trials ($n = 323$), n (%)	Non-published trials ($n = 131$), n (%)	P-value
Funding source ^a				
Industry	198 (44)	125 (39)	73 (56)	< 0.01
NIH	52 (11)	42 (13)	10 (8)	
US Federal	18 (4)	12 (4)	6 (5)	
Other	286 (63)	218 (67)	68 (52)	
Intervention ^a				
Drug	212 (47)	153 (47)	59 (45)	0.63
Behavioural	44 (10)	34 (11)	10 (8)	
Device	98 (22)	64 (20)	34 (26)	
Procedure	39 (9)	27 (8)	12 (9)	
Other	102 (22)	73 (23)	29 (22)	
Trial phase ^b				
Phase 1	28 (6)	13 (4)	15 (11)	< 0.01
Phase 2	107 (24)	84 (26)	23 (18)	
Phase 3	81 (18)	64 (20)	17 (13)	
Phase 4	46 (10)	31 (10)	15 (11)	
Not applicable	192 (42)	131 (41)	61 (47)	
Masking				
Open label	203 (45)	136 (42)	67 (51)	0.28
Single blind	57 (13)	41 (13)	16 (12)	
Double blind	61 (13)	45 (14)	16 (12)	
Triple blind	39 (9)	31 (10)	8 (6)	
Quadruple blind	91 (20)	69 (21)	22 (17)	
Unspecified	3 (1)	1 (0)	2 (2)	
Enrolment size				
Median (IQR)	62 (29.25–220.75)	80 (33–262)	43 (20–99.5)	
SD no. of participants	840	941	473	
<10	27 (6)	7 (2)	15 (11)	< 0.01
<;50	167 (37)	102 (32)	57 (44)	
<100	86 (19)	68 (21)	26 (20)	
<500	123 (27)	97 (30)	29 (22)	
>500	51 (11)	49 (15)	4 (3)	

Table 3 Characteristics of completed randomized trials registered with ClinicalTrials.gov

NIH, National Institutes of Health.

^aTrials with multiple funding sources or interventions are listed within all relevant categories; totals therefore add to more than 100%. ^bTrials described as 'phase I/II' in their registry entry are categorized as phase II. Trials described as 'phase II/III' in their registry entry are categorized as phase II.

Discussion

Our study highlights that discontinuation and non-publication of HF clinical trials is common (Figure 3). Around 21% of trials registered on ClinicalTrials.gov were discontinued prior to completion, and over a guarter of completed trials were never published. Our results are consistent with previous studies observing high non-publication and discontinuation rates across other medical specialties.9,10,13 These findings raise ethical concerns regarding the enrolment of human subjects and suggest that investigators and sponsors are not consistently fulfilling responsibilities to HF research participants. A total of 24 780 participants were enrolled in trials that were discontinued prior to completion, while 18 471 participants were enrolled in trials that never got published. This represents a large cohort of HF patients who subjected themselves to the potential risks, inconvenience, and discomfort of participating in a clinical trial without any contribution to the scientific literature, rendering their efforts futile.

Problems with recruitment of patients were cited as the most common reason for trial discontinuation. This may be of particular concern in HF trials that are considered more complex, and achieving adequate enrolment may require high cost and time and addition of enrolment sites.²⁰ This was seen in both Randomized Evaluation of Mechanical Assistance for the Treatment of Congestive Heart Failure (RE-MATCH) trial²¹ as well as Surgical Treatment for Ischemic Heart Failure (STICH) trial,²² where additional funding and global expansion in recruitment was required to meet the set enrolment targets.²⁰ Moreover, racial and socio-economic reasons might also serve as a potential barrier, whereby participants of different demographics may feel uncomfortable interacting with or trusting trial investigators or trial sponsors.²⁰ These factors may impair overall trial enrolment while also contributing to underrepresentation and impaired representativeness of racial/ethnic minorities and women across HF clinical trials.^{23–26}

Prior analyses have shown that high non-publication rates may be due to publication bias against studies that yield negative results.^{14,15,27} Considering that most of our trials involving non-publication were based on drug or device interventions, it may be possible that drugs showing low efficacy and negative outcomes remained unpublished.²⁸ This may be particularly true for HF trials where rates of trials meeting their primary endpoint have significantly decreased over time,¹⁶ potentially reducing the likelihood of trial results

FIGURE 3 The number of HF trials that were discontinued prematurely and have been unpublished. The most frequently cited reasons for premature discontinuation of HF trials and proposed strategies to decrease non-publication of HF clinical trials are also highlighted. RIAT, Restoring Invisible and Abandoned Trials.



being reported or subsequently published. Moreover, non-publication of negative outcomes is also observed majorly amongst industry funded studies as opposed to government funded trials.^{29,30} Furthermore, we did not limit our search with the number of participants enrolling in the trial, with most unpublished trials having less than 50 participants (44%). Publication bias is often noted in trials with small sample size, with an increased likelihood of publication associated with an increase in sample size.³¹ This relationship between low enrolment and non-publication may potentially further worsen enrolment difficulties, as patients may hesitate to participate in trials for fear of their efforts being worthless.

The emphasis by WHO on mandating access to results of all registered trials within 12 months is particularly welcomed to curtail information lost through unpublished trials.⁶ One way of practical implementation may be through audit, where a registry may be used to identify completed trials that remain unpublished even after a year of completion, post the date of results appearing in the registry, and derive performance metrics for individual companies, funding sources, interventions, institutes, and investigators.³² This information would account for greater transparency and ensure accountability.^{32,33} Second, RIAT (Restoring Invisible and Abandoned Trials) initiative should be considered. This initiative calls for investigators with unpublished data to signal their intention to publish their findings within a year or else provide public access to their data allowing independent investigators the ability to publish it.³⁴ Third, investigators publishing trial results on ClinicalTrials.gov website before publishing their findings in a journal should not be subjected to penalties,³⁵ which can potentially result in non-publication if a journal refuses to accept the manuscript. Outcomes from these studies should be publicly available without formal publication. Fourth, journal editors should adopt policies to encourage publication of all kinds of clinical trials, including trials that were terminated prematurely. For instance, to curb the pressing issue of non-publication in HF, a Dead Letter Office was initiated by Journal of American College of Cardiology - Heart Failure (JACC HF) as a repository of clinical trials that had previously remained unpublished.³⁶ Lastly, although termination of trials is warranted when interim analyses raise safety or efficacy concerns, it is crucial that results of these trials are also available, so that other investigators considering similar studies are aware of the risks associated with similar interventions.

While interpreting our results, there are a few limitations that should be considered. First, we only analysed trials that were listed in ClinicalTrials.gov. Second, we included the trial information as provided by the trial sponsors and investigators. Although registered trial data are verified by ClinicalTrials.gov staff for accuracy,³⁷ we were unable to verify data for all trials included in our analysis. Third, we only included reasons for discontinuation as listed in ClinicalTrials.

gov. In an event that there were multiple reasons for termination but only one listed, we could only record the one reason that was reported. Fourth, although a substantial amount of lag time was given, we cannot exclude the possibility of very late publication of trial results. For instance, the Dead Letter Office of JACC HF has published results of trials that were completed more than a decade ago.³⁶ Nevertheless, timely publication is important, and such delayed publication is still inconsistent with the investigator responsibility to trial participants. Finally, there were missing data for some variables, including reason for trial discontinuation.

In conclusion, we observed that HF clinical trials are frequently discontinued and may remain unpublished. This raises ethical concerns towards trial participants and the potential risks, inconvenience, and discomfort these individuals expose themselves to, while contributing nothing to the medical literature. Results should explicitly be available for all registered trials, including the ones terminated early or due to safety or toxicity reasons to ensure complete transparency in clinical trial reporting. The publication bias associated with these clinical trials needs to be mitigated, and additional policies involving greater accountability for those responsible for unpublished trials should be implemented to protect those participating in HF trials from having their efforts result in dissemination of no incremental scientific knowledge.

Conflict of interest

S.J.G. is supported by the National Heart Lung and Blood Institute T32 postdoctoral training grant (T32HL069749-14) and a Heart Failure Society of America/Emergency Medicine Foundation Acute Heart Failure Young Investigator Award funded by Novartis, has received research support from Amgen, Bristol-Myers Squibb, and Novartis, and serves on an advisory board for Amgen. M.M. declares that he has received research grant support from Amgen and consulting fees from Amgen, Bayer, Fresenius, Novartis, Servier, Vifor, and Windtree Therapeutics. S.D.A. has received research support from Vifor International 38; Abbott Vascular, and fees for consultancy and/or speaking from Astra-Zeneca, Bayer, Boehringer Ingelheim, Respicardia, Impulse Dynamics, Janssen, Novartis, Servier, and Vifor International. G.F. participated in committees of trials and registries sponsored by BI, Bayer, Medtronic, Servier, Novartis, Vifor and Agmen. G.C.F. reports consulting for Abbott, Amgen, Bayer, Janssen, Medtronic, and Novartis. J.B. is a consultant for Abbott, Amgen, Applied Therapeutics, Astra Zeneca, Bayer, Boehringer Ingelheim, Bristol Myers Squib, CVRx, Janssen, LivaNova, Luitpold, Medtronic, Merck, Novartis, Relypsa, and Vifor. All other authors report no disclosures.

References

- De Angelis C, Drazen JM, Frizelle FA, Haug C, Hoey J, Horton R, Kotzin S, Laine C, Marusic A, Overbeke AJ, Schroeder TV, Sox HC, Van Der Weyden MB. Clinical trial registration: a statement from the *International Committee* of *Medical Journal. Lancet* 2004; 364: 911–912.
- Moreno J, Caplan AL, Wolpe PR. Updating protections for human subjects involved in research. Project on Informed Consent, Human Research Ethics Group. JAMA 1998; 280: 1951–1958.
- Thornton A, Lee P. Publication bias in meta-analysis: its causes and consequences. J Clin Epidemiol 2000; 53: 207–216.
- 4. Rising K, Bacchetti P, Bero L. Reporting bias in drug trials submitted to the Food and Drug Administration: review of publication and presentation. *PLoS Med* 2008; **5**: e217.
- Chan AW, Song F, Vickers A, Jefferson T, Dickersin K, Gotzsche PC, Krumholz HM, Ghersi D, van der Worp HB. Increasing value and reducing waste: addressing inaccessible research. *Lancet* 2014; **383**: 257–266.
- World Health Organisation. WHO statement on public disclosure of clinical trial results. https://www.who.int/ictrp/results/reporting/en/ (30 October 2020).
- U.S Food and Drug Administration. FDA00027;s role: ClinicalTrials.gov Information. https://www.fda.gov/science-research/clinical-trials-and-human-subject-protection/fdas-roleclinicaltrialsgov-information (30 October 2020).
- Ioannidis JP. Effect of the statistical significance of results on the time to completion and publication of randomized efficacy trials. *JAMA* 1998; 279: 281–286.
- Jones CW, Handler L, Crowell KE, Keil LG, Weaver MA, Platts-Mills TF. Nonpublication of large randomized clinical trials: cross sectional analysis. *BMJ* 2013; 347: f6104.
- Pica N, Bourgeois F. Discontinuation and nonpublication of randomized clinical trials conducted in children. *Pediatrics* 2016; **138**: e20160223.
- Yilmaz T, Jutten RJ, Santos CY, Hernandez KA, Snyder PJ. Discontinuation and nonpublication of interventional clinical trials conducted in patients with mild cognitive impairment and Alzheimer00027;s disease. *Alzheimers Dement (N Y)* 2018; 4: 161–164.
- Kasenda B, von Elm E, You J, Blumle A, Tomonaga Y, Saccilotto R, Amstutz A, Bengough T, Meerpohl JJ, Stegert M, Tikkinen KA, Neumann I, Carrasco-Labra A, Faulhaber M, Mulla SM, Mertz D, Akl EA, Bassler D, Busse JW,

Ferreira-Gonzalez I, Lamontagne F, Nordmann A, Gloy V, Raatz H, Moja L, Rosenthal R, Ebrahim S, Schandelmaier S, Xin S, Vandvik PO, Johnston BC, Walter MA, Burnand B, Schwenkglenks M, Hemkens LG, Bucher HC, Guyatt GH, Briel M. Prevalence, characteristics, and publication of discontinued randomized trials. *JAMA* 2014; **311**: 1045–1051.

- Roddick AJ, Chan FTS, Stefaniak JD, Zheng SL. Discontinuation and non-publication of clinical trials in cardiovascular medicine. *Int J Cardiol* 2017; 244: 309–315.
- Hopewell S, Loudon K, Clarke MJ, Oxman AD, Dickersin K. Publication bias in clinical trials due to statistical significance or direction of trial results. *Cochrane Database Syst Rev* 2009; (1): Mr000006.
- Duyx B, Urlings MJE, Swaen GMH, Bouter LM, Zeegers MP. Scientific citations favor positive results: a systematic review and meta-analysis. J Clin Epidemiol 2017; 88: 92–101.
- Samman Tahhan A, Vaduganathan M, Kelkar A, Georgiopoulou VV, Kalogeropoulos AP, Greene SJ, Fonarow GC, Gheorghiade M, Butler J. Trends in heart failure clinical trials from 2001– 2012. J Card Fail 2016; 22: 171–179.
- ClinicalTrials.gov. ClinicalTrials.gov Background. https://clinicaltrials.gov/ ct2/about-site/background (30 October 2020).
- US National Library of Medicine. FDAAA 801 and the Final Rule. https://www. clinicaltrials.gov/ct2/manage-recs/ fdaaa (30 October 2020).
- Ross JS, Mocanu M, Lampropulos JF, Tse T, Krumholz HM. Time to publication among completed clinical trials. JAMA Intern Med 2013; 173: 825–828.
- 20. Weisfeld VER, Claiborne A. Recruitment challenges in clinical trials for different diseases and conditions. In *Public Engagement and Clinical Trials: New Models and Disruptive Technologies*. National Academy of Sciences; 2012.
- Rose EA, Gelijns AC, Moskowitz AJ, Heitjan DF, Stevenson LW, Dembitsky W, Long JW, Ascheim DD, Tierney AR, Levitan RG, Watson JT, Meier P, Ronan NS, Shapiro PA, Lazar RM, Miller LW, Gupta L, Frazier OH, Desvigne-Nickens P, Oz MC, Poirier VL. Long-term use of a left ventricular assist device for end-stage heart failure. N Engl J Med 2001; 345: 1435–1443.
- 22. Velazquez EJ, Lee KL, Deja MA, Jain A, Sopko G, Marchenko A, Ali IS, Pohost G, Gradinac S, Abraham WT, Yii M, Prabhakaran D, Szwed H, Ferrazzi P, Petrie MC, O00027;Connor CM, Panchavinnin P, She L, Bonow RO, Rankin GR, Jones RH, Rouleau JL. Coronary-artery bypass surgery in patients

with left ventricular dysfunction. *N Engl J Med* 2011; **364**: 1607–1616.

- Khan MS, Shahid I, Siddiqi TJ, Khan SU, Warraich HJ, Greene SJ, Butler J, Michos ED. Ten-Year Trends in Enrollment of Women and Minorities in Pivotal Trials Supporting Recent US Food and Drug Administration Approval of Novel Cardiometabolic Drugs. Journal of the American Heart Association 2020; 9. https://doi.org/ 10.1161/jaha.119.015594
- 24. Tahhan AS, Vaduganathan M, Greene SJ, Fonarow GC, Fiuzat M, Jessup M, Lindenfeld J, O00027;Connor CM, Butler J. Enrollment of older patients, women, and racial and ethnic minorities in contemporary heart failure clinical trials: a systematic review. JAMA Cardiol 2018; 3: 1011–1019.
- 25. Greene SJ, DeVore AD, Sheng S, Fonarow GC, Butler J, Califf RM, Hernandez AF, Matsouaka RA, Samman Tahhan A, Thomas KL, Vaduganathan M, Yancy CW, Peterson ED, 000027; Connor CM, Mentz RJ. Representativeness of a heart failure trial by race and sex: results from ASCEND-HF and GWTG-HF. JACC Heart Fail 2019; 7: 980–992.
- Levinsson A, Dubé MP, Tardif JC, de Denus S. Sex, drugs, and heart failure: a sex-sensitive review of the evidence base behind current heart failure clinical guidelines. *ESC Heart Fail* 2018; 5: 745–754.
- Turner EH, Matthews AM, Linardatos E, Tell RA, Rosenthal R. Selective publication of antidepressant trials and its influence on apparent efficacy. *N Engl J Med* 2008; **358**: 252–260.
- Hauck WW, Anderson S. A proposal for interpreting and reporting negative studies. *Stat Med* 1986; 5: 203–209.
- Davidson RA. Source of funding and outcome of clinical trials. J Gen Intern Med 1986; 1: 155–158.
- 30. Yaphe J, Edman R, Knishkowy B, Herman J. The association between funding by commercial interests and study outcome in randomized controlled drug trials. *Fam Pract* 2001; **18**: 565–568.
- Easterbrook PJ, Berlin JA, Gopalan R, Matthews DR. Publication bias in clinical research. *Lancet* 1991; 337: 867–872.
- 32. Goldacre B. How to get all trials reported: audit, better data, and individual accountability. *PLoS Med* 2015; **12**: e1001821.
- 33. Ponikowski P, Anker SD, AlHabib KF, Cowie MR, Force TL, Hu S, Jaarsma T, Krum H, Rastogi V, Rohde LE, Samal UC, Shimokawa H, Budi Siswanto B, Sliwa K, Filippatos G. Heart failure: preventing disease and death worldwide. ESC Heart Fail 2014; 1: 4–25.

- Doshi P, Dickersin K, Healy D, Vedula SS, Jefferson T. Restoring invisible and abandoned trials: a call for people to publish the findings. *BMJ* 2013; 346: f2865.
- O'Connor CM. Why should we publish all clinical trials? *JACC Heart Fail* 2019; 7: 730.
- 36. O'Connor C. Dead letter office. JACC Heart Fail 2013; 1: 181–182.
- 37. US National Library of Medicine. ClinicalTrials.gov and related projects: improving access to information about clinical studies In; 2013.